Author's response to reviews

Title: Practice Nurses’ views of their role in the management of Chronic Fatigue Syndrome/Myalgic Encephalitis: a qualitative study

Authors:

Carolyn A Chew-Graham (cchew@manchester.ac.uk)
Rebecca Dixon (Rebecca.Dixon@student.manchester.ac.uk)
Jonathan W Shaw (jonathanwilliamshaw@student.manchester.ac.uk)
Nina Smyth (Nina.Smyth@student.manchester.ac.uk)
Karina Lovell (karina.lovell@manchester.ac.uk)
Sarah Peters (Sarah.Peters@manchester.ac.uk)

Version: 5 Date: 11 January 2009

Author’s response to reviews:

Dear Editor

The major revision has been carried out - the table has been revised and now has no vertical lines.

The minor revisions have been carried out (except removal of qualifications from manuscript and none were included in the previous version.

I hope this paper is now acceptable

Yours sincerely

Carolyn Chew-Graham (corresponding author)